Author's response to reviews

Title: A systematic review of mental disorder, suicide, and deliberate self harm in lesbian, gay and bisexual people

Authors:

Michael King (m.king@medsch.ucl.ac.uk)
Joanna Semlyen (josemlyen@hotmail.com)
Sharon See Tai See Tai (s.seetai@pcps.ucl.ac.uk)
Helen Killaspy (h.killaspy@medsch.ucl.ac.uk)
David Osborn (d.osborn@medsch.ucl.ac.uk)
Dmitri Popelyuk (d_popelyuk@hotmail.com)
Irwin Nazareth (i.nazareth@pcps.ucl.ac.uk)

Version: 4 Date: 5 August 2008

Author's response to reviews:

Thank you for your email of 28th July. The revision contains the following corrections/changes:

1. The title has been made clearer.
2. References have been cited in the text using consecutive numbers in square brackets.
3. Competing interest statement added.
4. Figure 10 in the text was an error. There are only 9 figures and this has been corrected.
5. Authors' qualifications, revision date and word count have been removed from title page. Email addresses have been added.
6. All coloured text in manuscript has been blackened. All coloured text in tables has been emboldened.
7. All underlining has been removed.
8. Authors’ contribution section has been edited.
9. Reference list edited to remove et al. and authors’ names added.